Author's response to reviews

Title: Socioeconomic factors and low birth weight in Mexico

Authors:

Laura P Torres-Arreola MD, MSc (laura.torres@imss.gob.mx)
Patricia Constantino-Casas MD, MEc (norma.constantino@imss.gob.mx)
Sergio Flores MD, MSc (sloresh@cis.gob.mx)
Juan P Villa-Barragan MD, MPH (villaj@mex.ops-oms.org)
Enrique Rendon MD, MSc (enrique.rendon@imss.gob.mx)

Version: 5 Date: 30 December 2004

Author's response to reviews: see over
Shingai Feresu on 11 November 2004

1. In Abstract section: the word “semester” was changed to “half”. In Introduction section: the comment about new literature showing that malnutrition and preconceptual weight are big factor in preterm, was taken into account. The suggested reference was reviewed and included.